Author's response to reviews

Title: Incidence and prevalence of upper-extremity musculoskeletal disorders. A systematic appraisal of the literature.

Authors:

Bionka MA Huisstede (b.huisstede@erasmusmc.nl)
Sita MA Bierma-Zeinstra (s.bierma-zeinstra@erasmusmc.nl)
Bart W Koes (b.koes@erasmusmc.nl)
Jan AN Verhaar (J.verhaar@erasmusmc.nl)

Version: 4 Date: 19 December 2005

Author's response to reviews: see over
Concerning: Formatting Changes for MS: [1594024931788483]

Dear Mister Austin,

Thank you for approving my manuscript “Incidence and prevalence of upper-extremity musculoskeletal disorders. A systematic appraisal of the literature” (MS: [1594024931788483]).

In my final manuscript I made the following changes:

1. In the abstract, results: … 13 studies were included.
2. References: ref.10 and ref. 11: I changed the layout.

Furthermore, I tried to change reference 18, but (by checking the 40 references) I did not succeed.

I’m looking forward to see the manuscript published.

Yours sincerely, on behalf of all authors,

Bionka Huisstede

Department of General Practice, Room Ff 325
Erasmus Medical Center, P.O. Box 1738, 3000 DR Rotterdam, The Netherlands